Author's response to reviews

Title: Are associations between electronic media use and BMI different across levels of physical activity? A cross-national investigation in 30 countries

Authors:

Ole Melkevik (ole.melkevik@fhi.no)
Ellen Haug (EllenMerete.Haug@NLA.no)
Mette Rasmussen (mera@niph.dk)
Anne Siri Fisman (anne-siri.fisman@iuh.uib.no)
Bente Wold (bente.wold@iuh.uib.no)
Alberto Borraccino (alberto.borraccino@hbsc.org)
Erik Sigmund (erik.sigmund@gmail.com)
Robert Balazsi (robertbalazsi@psychology.ro)
Jens Bucksch (jens.bucksch@uni-bielefeld.de)
Jo Inchley (jci2@st-andrews.ac.uk)
Margarida Gaspar de Matos (margaridagaspar@netcabo.pt)
Oddrun Samdal (oddrun.samdal@rektor.uib.no)

Version: 5 Date: 28 April 2015

Author's response to reviews:

Dear editor

We have now provided a new authors contributions section as requested.

I am very sorry for the substantial delay in response!
Please let us know if there are any other changes you would like us to do.

on behalf of the authors
Ole Melkevik